Author's response to reviews

Title: Knowledge Translation Research in Population Health: Establishing a Collaborative Research Agenda

Authors:

Christian Dagenais (christian.dagenais@umontreal.ca)
Valéry Ridde (valery.ridde@umontreal.ca)
Marie-Claire Laurendeau (Marie-Claire.Laurendeau@inspq.qc.ca)
Karine Souffez (Karine.Souffez@inspq.qc.ca)

Version: 2 Date: 1 September 2009

Author's response to reviews:

All the changes asked by the two reviewers have been made. These changes are highlighted in the text. We would like to thank the reviewers for their very useful comments on the first version of the manuscript. We hope that the changes made will be satisfactory.